Author's response to reviews


Authors:
  Paul A Batchelor (paulb@public-health.ucl.ac.uk)
  Aubrey Sheiham (a.sheiham@ucl.ac.uk)

Version: 4 Date: 14 December 2005

Author's response to reviews: see over
We have altered the shortcomings in grammar identified by the second referee and modified the text to ensure consistency in the usage of various phrases. In addition we have made the corrections to the spacing within the references as well as updating the reference as suggested along with correcting the order of references 6 and 7.